Author’s response to reviews

Title: Solid Food Protein induced Enterocolitis Syndrome - four new cases, a brief review of the literature and possible new insights of the pathological mechanisms

Authors:

Amolak S Bansal (Amolak.Bansal@esth.nhs.uk)
Sree Bhaskaran (Sree.Bhaskaran@esth.nhs.uk)
Rhea A Bansal (RheaBansal@yahoo.co.uk)

Version: 2 Date: 15 January 2012

Author's response to reviews:

All the changes necessary for submission to the Journal of Medical Case reports have now been made.